\begin{table}%t4
\caption{\label{tab:CEMP-binaries}Periods and eccentricities of CEMP~stars in binaries from our selection ($[{\rm Fe}/{\rm H}]=-2.3$~$\pm$ $0.5$, $\log_{10}g\leq4$).}
\par
\small%\centerline
 {
\begin{tabular}{cccc}
\hline\hline \noalign{\smallskip}
Object & Period & $e$ & Reference\\
\hline
CS 22948-027 & 505d & 0.3 & {\citet{2001AJ....122.1545P}}\\
CS 22948-027 & 426.5d & 0.02  & {\citet{2005AA...429.1031B}}\\
HD 5223 & 755.2d & 0 & {\citet{1990mcclure_woodsworth}}\\
HD 224959 & 1273d & 0.179 & {\citet{1990mcclure_woodsworth}}\\
CS 22956-028 & 1290d & 0.22 & {\citet{2003ApJ...592..504S}}\\
CS 22942-019 & 2800d & 0.19 & {\citet{2001AJ....122.1545P}}\\
LP 625-44 & 12y &  & {\citet{2000ApJ...536L..97A}}\\
\hline
\end{tabular}}
\medskip
Note that two solutions are available for CS~22948-027.
\end{table}